Author's response to reviews

Title: Smorgasbord or Symphony? Assessing public health nutrition policies across 30 European countries using a novel framework: A mixed methods study

Authors:

Ffion Lloyd-Williams (ffionlw@liv.ac.uk)
Helen Bromley (bromley@liv.ac.uk)
Lois Orton (lorton@liv.ac.uk)
Corinna Hawkes (c.hawkes@wcrf.org)
David Taylor-Robinson (dctr@liv.ac.uk)
Martin O'Flaherty (moflaher@liv.ac.uk)
Rory McGill (rmcgill@liv.ac.uk)
Elspeth Anwar (anware@liv.ac.uk)
Lirije Hyseni (hysen001@liv.ac.uk)
May Moonan (mmoonan@liv.ac.uk)
Mike Rayner (mike.rayner@dph.ox.ac.uk)
Simon Capewell (capewell@liv.ac.uk)

Version: 4 Date: 7 October 2014

Author's response to reviews: see over
Dr. Natalie Pafitis,
Executive Editor
BMC Public Health

Dear Dr. Pafitis

Smorgasbord or Symphony? Assessing public health nutrition policies across 30 European countries using a novel framework: A mixed methods study

We have amended the manuscript based upon the Editorial Requests below:

(1) A statement in the Methods section of the manuscript body (line 256), confirming we have adhered to the RATS guidelines for reporting qualitative studies.

(2) The abstract has been shortened to 347 words, according to BioMed Central guidelines.

We look forward to hearing from you.

Yours sincerely

Dr. Ffion Lloyd-Williams,
on behalf of coauthors